Author's response to reviews

Title: Diagnostic performance of contrast enhanced CT and 18F-FDG PET/CT in suspicious recurrence of biliary tract cancer after curative resection

Authors:

Yun-Gyoo Lee (gosciny@gmail.com)
Sae-Won Han (saewonhan@medimail.co.kr)
Do-Youn Oh (phdoyoun@snu.ac.kr)
Eui Kyu Chie (ekchie93@snu.ac.kr)
Jin-Young Jang (jangjy4@snu.ac.kr)
Seock-Ah Im (moisa@snu.ac.kr)
Tae-You Kim (kimty@snu.ac.kr)
Sun-Whe Kim (sunkim@snu.ac.kr)
Sung Whan Ha (swha@snu.ac.kr)
Yung-Jue Bang (bangyj@snu.ac.kr)

Version: 4 Date: 19 May 2011

Author's response to reviews:

Dear Editor-in Chief of BioMed Central Cancer

We deeply appreciate your careful comments for our manuscript entitled, “Diagnostic performance of contrast enhanced CT and 18F-FDG PET/CT in suspicious recurrence of biliary tract cancer after curative resection”. We have revised our manuscript regarding to editors' comments.

Thank you again for giving us important comments on our work.

Sincerely,

Do-Youn Oh, M.D., Ph.D.

Assistant Professor

Department of Internal Medicine, Seoul National University Hospital, Seoul, Republic of Korea

# Major revisions

1. Affiliations

We repeated institutional addresses of affiliations 1,2 and 3 in full. (Page 1, line 6-10)

2. Tables
We included the tables in the manuscript in a tables section following the references. (Page 23-28) Then, we removed the additional file including the tables.

3. Figures

We removed the figure details from the 'additional files' section.

# Minor revisions

1. Previous exhibition

We moved the information about previous exhibition to the 'Acknowledgements' section.

   (Page 19, line 16-17)

2. Colored text

We changed all colored text to black.

3. Author list

We removed qualifications from author list. (Page 1. line 4-5)